Author’s response to reviews

Title: Coexistent papillary thyroid carcinoma diagnosed in surgically treated patients for primary versus secondary hyperparathyroidism: same incidence, different characteristics

Authors:

Cristina Preda (cpredal@yahoo.com)
Dumitru Branisteanu (dumitru_branisteanu@yahoo.com)
Ioana Armasu (ioana.armasu@umfiasi.ro)
Rau Danila (r_danila@yahoo.com)
Cristian Velicescu (cristianvelicescu@yahoo.com)
Delia Ciobanu (deliaku@yahoo.com)
Adrian Covic (adrian.covic@umfiasi.ro)
Alexandru Grigorovici (alexandrugrigorovici@yahoo.com)

Version: 3 Date: 03 Jul 2019

Author’s response to reviews:

Dear Dr. Andrea Polistena,

Thank you for your kind advice. We performed the required changes in text and loaded the revised manuscript.

Hoping that the manuscript fulfills now the Journal's requests, please do not hesitate however to contact us if other changes are needed,

Sincerely Yours,

Dumitru D. Branisteanu
Corresponding Author